Author's response to reviews

Title: Short-term Sick Leave and Future Risk of Sickness Absence and Unemployment - the Impact of Health Status

Authors:

Christina Lindholm (christina.lindholm@ki.se)
Hanna Hultin (hanna.hultin@ki.se)
Mauricio Malfert (mauricio.malfert@ucr.uu.se)
Jette Möller (jette.moller@ki.se)

Version: 2 Date: 16 December 2011

Author's response to reviews:

Dear editor,

We have made some minor changes to the manuscript, changing a couple of words that were marked in red by the word processing program and refitting table 3, which did now show properly in the previous version.

We hope you consider these changes when making your initial assessment.

Kind regards,
Hanna Hultin
on behalf of all authors